Digital interventions for hypertension and asthma to support patient self-management in primary care: the DIPSS research programme including two RCTs

Lucy Yardley,^{1,2*} Kate Morton,¹ Kate Greenwell,¹ Beth Stuart,³ Cathy Rice,⁴ Katherine Bradbury,¹ Ben Ainsworth,¹ Rebecca Band,⁵ Elizabeth Murray,⁶ Frances Mair,⁷ Carl May,⁸ Susan Michie,⁹ Samantha Richards-Hall,⁴ Peter Smith,¹⁰ Anne Bruton,⁵ James Raftery,³ Shihua Zhu,³ Mike Thomas,³ Richard J McManus¹¹ and Paul Little³

- ¹School of Psychology, University of Southampton, Southampton, UK
- ²School of Psychological Science, University of Bristol, Bristol, UK
- ³Primary Care, Population Sciences and Medical Education, University of Southampton, Southampton, UK
- ⁴Patient and public involvement contributor, UK
- ⁵School of Health Sciences, Faculty of Environmental and Life Sciences, University of Southampton, Southampton, UK
- ⁶Primary Care and Population Health, University College London, London, UK
- ⁷General Practice and Primary Care, University of Glasgow, Glasgow, UK
- ⁸Faculty of Public Health and Policy, London School of Hygiene and Tropical Medicine, London, UK
- ⁹Centre for Behaviour Change, Research Department of Clinical, Educational and Health Psychology, University College London, London, UK
- ¹⁰Department of Social Statistics and Demography, University of Southampton, Southampton, UK
- ¹¹Nuffield Department of Primary Care Health Sciences, University of Oxford, Oxford, UK

*Corresponding author lucy.yardley@bristol.ac.uk

Declared competing interests of authors: Lucy Yardley reports membership of the Health Technology Assessment (HTA) Antimicrobial Resistance Board (2013–14), HTA Efficient Study Designs Board (2015–16) and Public Health Research Funding Board (2015–17). Beth Stuart reports membership of the HTA Commissioning Committee (2020–4). Ben Ainsworth reports membership of the HTA Commissioning Committee (2020–1). Elizabeth Murray received grants from the Wellcome Trust India Alliance (Hyderabad, India), the National Health and Medical Research Council (Canberra, ACT, Australia), Alcohol Research UK (London, UK), Macmillan Cancer Support (London, UK) and the Medical Research Council (London, UK), during the conduct of the study. Frances Mair received personal fees from Janssen-Cilag Limited for activities outside the submitted work. James Raftery is an active member of the National Institute for Health and Care Research (NIHR) HTA Editorial Board and the NIHR Efficacy and Mechanism Evaluation Editorial Board. Mike Thomas reports membership of the HTA PCCPI Panel (2015–18). Richard J McManus received blood pressure monitors for research purposes from OMRON (Milton Keynes, UK), grants from the Stroke Association (London, UK) and travel funding from the European Society of Hypertension, outside the submitted work. In addition, Richard J McManus reports membership of the HTA Clinical Evaluation and Trials Committee (2010–15). Paul Little was the director of NIHR Programme Grants for Applied Research programme and reports membership of the NIHR Journals Library Board (2012–18).

Published December 2022 DOI: 10.3310/BWFI7321

Plain English summary

DIPSS research programme including two RCTs Programme Grants for Applied Research 2022; Vol. 10: No. 11 DOI: 10.3310/BWFI7321

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

L ong-term conditions can be difficult and costly to manage. Online interventions (e.g. websites) can support people to look after their health at home, but we need to understand how to make these online interventions acceptable and effective.

We carried out a review of existing research, which showed that digital interventions could lower blood pressure and improve asthma symptoms, but the evidence was varied in terms of how well the interventions worked. We also developed and evaluated two online interventions (one for high blood pressure and one for asthma). Detailed feedback from patients and general practitioners helped us to improve the interventions to ensure that they were persuasive and easy to understand.

Our hypertension intervention (i.e. HOME BP) helped patients to monitor their own blood pressure at home and prompted general practitioners to change medication when the patient's blood pressure was raised over time. A trial with 622 patients found that after 1 year patients using the HOME BP intervention had lower blood pressure than patients receiving usual care. The HOME BP intervention had a high probability of being cost-effective in relation to the criteria used by the NHS.

Our asthma intervention (i.e. My Breathing Matters) provided information and support to help patients engage in activities that would help them to better control their asthma. For example, using their medication as prescribed or learning breathing exercises. We carried out a small trial to check whether or not our research procedures were feasible. We recruited 88 asthma patients (our target was 80 patients) and only a small number of patients did not complete questionnaires at all time points, suggesting that it would be worthwhile testing the asthma intervention with a larger number of people.

Interviews with patients and general practitioners suggested that the online interventions were acceptable and useful for helping to manage high blood pressure and asthma. This research suggested modifications for improving users' experiences.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

Programme Grants for Applied Research (PGfAR) was launched in 2013 and is indexed by Europe PMC, NCBI Bookshelf, DOAJ, Ulrichsweb[™] (ProQuest LLC, Ann Arbor, MI, USA) and Scopus[®] (Elsevier, Amsterdam, Netherlands).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar.

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health and Care Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-1211-20001. The contractual start date was in January 2014. The final report began editorial review in August 2019 and was accepted for publication in March 2022. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the PGfAR programme or the Department of Health and Social Care.

Copyright © 2022 Yardley *et al.* This work was produced by Yardley *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Dr Cat Chatfield Director of Health Services Research UK

NIHR Journals Library Editors

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Interim Chair of HTA and EME Editorial Board. Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Rob Riemsma Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk